



# Article The Moderating Effect of a Golden Parachute on the Association between CSR and Firm Value: Does Gender-Driven Innovation Matter?

Collins E. Okafor <sup>1</sup>, Nacasius U. Ujah <sup>2</sup>, Eunho Cho <sup>1</sup>, \*, Winifred U. Okafor <sup>3</sup> and Kevin L. James <sup>1</sup>

- <sup>1</sup> Department of Accounting and Finance, North Carolina A&T State University, Greensboro, NC 27411, USA
- <sup>2</sup> Ness School of Management and Economics, South Dakota State University, Brookings, SD 57006, USA
  - <sup>3</sup> University of Texas at San Antonio, San Antonio, TX 78249, USA
  - \* Correspondence: echo@ncat.edu

Abstract: We revisit the debate on whether a firm's corporate social responsibility (CSR) activities enhance firm value. Research on related topics has produced mixed results suggesting a need to further investigate factors that directly or indirectly impact the CSR-firm value association. To this end, we examine if a firm's adoption of a golden parachute (GP) moderates the relationship between CSR and firm value. We also investigate if diversity-based innovation as it pertains to the gender of executives reveals any difference in the CSR-firm value relation. Using a sample of 11,065 firm-year observations of publicly traded US firms from 2007 to 2016, we find that CSR activities are significantly and positively associated with firm value. More importantly, our study shows that for US firms that issue GPs, this severance pay strengthens this positive relationship, suggesting that CEOs with a GP engage in more value-enhancing innovative CSR projects than their counterparts without it. This finding supports the conflict resolution theory and the resource-based view of the firm. A test to examine if the gender of the corporate executives alters their behavior towards CSR when the GP protects them shows an inverse relationship between female executives and CSR-firm value association. This interesting finding lends credence to related theories suggesting that women in male-dominated fields may feel pressured to conform to the stereotype of women as less competent than men and may adopt traditionally masculine behaviors to counteract this stereotype. As they climb the corporate leadership ladder endowed with a GP, the stereotype threat may still prevail, adversely affecting the CSR-firm value outcomes. These results remain robust after a series of sensitivity tests.

**Keywords:** corporate social responsibility; executive compensation; golden parachutes; firm value; innovation

# 1. Introduction

Corporate social responsibility (CSR) is a crucial component of contemporary business, and innovative CSR strategies can help organizations stand out from the competition and positively impact society. However, there is no consensus in the literature on the impact of CSR on firm performance. There is evidence suggesting a positive association between a firm's CSR engagements and its financial performance [1–4]. In contrast, some empirical studies find a negative association [5–7]. Others see no association [8,9].

CSR is defined as the actions that a firm chooses to take that substantially enhance the well-being of its stakeholders, community, and environment [10]. The nonconsensus enumerated above has triggered skepticism surrounding CSR's value-enhancing potential. It creates a void and gap in the literature, clogging the wheels of the CSR campaign. It has compelled some to posit that CSR is antithetical to sound business practice because it dilutes the firm's focus on the shareholders' wealth creation [11,12]. However, advocates of CSR argue that it is vital for sustainable business operations. Thus, firms must look beyond



Citation: Okafor, C.E.; Ujah, N.U.; Cho, E.; Okafor, W.U.; James, K.L. The Moderating Effect of a Golden Parachute on the Association between CSR and Firm Value: Does Gender-Driven Innovation Matter? *Sustainability* **2023**, *15*, 5483. https://doi.org/10.3390/su15065483

Academic Editor: Wen-Hsien Tsai

Received: 15 February 2023 Revised: 16 March 2023 Accepted: 19 March 2023 Published: 20 March 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). narrow economic returns and embrace CSR's broader social impact [13]. CSR is paramount to society. The lack of unanimity in the related literature on the impact of CSR on firm value calls for further investigation examining how firms can better strategically align the interests of both the investing and non-investing stakeholders. This compelling need is a motivating factor that drives our research quest.

This study contributes to the literature on similar topics investigating the association between CSR and firm value. Focusing on the US corporate environment, we extend prior literature by examining CEOs' behavior toward CSR when they have a golden parachute (GP) (A golden parachute is a contractual clause in the executives' compensation package that guarantees a set of benefits to a CEO if a firm is acquired, employment is terminated, or the CEO remains with the firm through a recessionary cycle.). The effect of the executives' gender on this relationship is also closely examined. Firms issue golden parachutes to attract the best talent for managerial positions and stay competitive in their respective industry. The uncertainty surrounding the value-adding potential of CSR poses some risk to both the firm and the executives. Is the manager protected from these risks? A golden parachute clause would somewhat insulate the manager from possible employment risks associated with the uncertainties of CSR investments. The provision of a golden parachute in the executive compensation contract requires that substantial benefits be given to the manager if their employment is terminated. The manager could become more willing to explore and promote innovative CSR investments with such a condition.

Corporate executives want to maximize the value of their golden parachutes. To achieve this, they cautiously and consciously approve those innovative projects with the highest likelihood of enhancing the firm value that, in return, should maximize the value of the golden parachute (According to Bloomberg, the largest portion of severance often comes from accelerated payments of unvested stock options and other stock units that would otherwise not be eligible for a payout—http://www.bloomberg.com/news/2013 \$-\$06\$-\$06/golden-parachutes (accessed on 12 March 2023)). One can logically assume that this category of executives would only pursue and approve CSR projects that are undoubtedly value-enhancing.

Prejudice and gender bias can also stifle innovation in the workplace. Women may encounter barriers at work, for instance, that restrict their ability to take the lead on or participate in innovation projects [14]. Women may also be less likely to be acknowledged or given credit for their creative suggestions, which deters them from making further contributions [15]. What happens when the GP protects women in top leadership positions? How does this affect their behavior towards CSR?

Our findings show that the association between CSR and firm performance is positive. First, we test for the moderation role of the golden parachute. The results show that it magnifies this positive association because the CEOs with this clause in their contract engage in more value-enhancing CSR projects than their counterparts without it. Our results support the conflict resolution theory and the resource-based view of CSR. These related theories argue that top managers use CSR as a device to develop and build good relationships with the firm's stakeholders, which ultimately leads to the enhancement of the firm value.

Second, we disaggregate the CSR measure to examine which types of CSR CEOs with golden parachutes believe are most likely to enhance firm value. The results of our analysis show that the golden parachutes positively moderate the value-adding power of three CSR components: community, human rights, and product.

Third, of distinct interest to this research, we classify the data into female and male groups. The results of the analysis conducted reveal that for firms with female executives who are protected by the GP, the relation between CSR and firm value is negative but not statistically significant, whereas for the male group, the strong positive moderating power of the GP is sustained. This contrasting finding could be explained by related theories on why women in male-dominated fields may feel pressure to adopt traditionally masculine behaviors, especially on their rise to the top of the corporate ladder.

Self-verification and social role theories suggest that women in male-dominated fields may feel pressure to adopt traditionally masculine behaviors because of societal expectations, the threat of being judged based on gender stereotypes, and a desire to maintain their sense of self. Drawing from the lens of stereotype threat theory, in the context of our study, the stigmatization of women in the corporate arena may cause them to experience anxiety or stress when they feel their performance could confirm a negative stereotype about their abilities. This anxiety or stress can impair their cognitive functioning and negatively impact their performance in managerial investment decisions that impact the CSR–firm value outcomes.

The rest of this study proceeds as follows: Section 2 provides the relevant literature review. Section 3 presents hypothesis development. Section 4 provides data and the research model. Section 5 presents the result analysis and endogeneity tests. Section 6 shows the discussion. Section 7 concludes the study.

# 2. Literature Review

#### 2.1. CSR and Golden Parachutes

The literature on CSR and firm value has produced mixed and sometimes conflicting findings. Empirical evidence supporting a positive association between CSR and firm value argues that CSR could enhance a firm's value in multiple ways. It could improve the firm's reputation and reduce the frequency of investors' explicit claims [16], reduce the cost of capital [17], and reduce an undesirable debt ratio [18]. There is also evidence to suggest that CSR increases the willingness of customers (investors) to pay premium prices for socially responsible products of the firm because a good reputation is usually considered a signal of quality products [19].

In contrast to the positive evidence enumerated above, there is also opposing evidence discrediting the positive association between CSR and firm performance. Some empirical studies find a negative association [5–7], and others see no association [8,9]. Reference [20] uses a meta-analysis of CSR and firms' performance to show the literature's ambiguity and inconsistent empirical findings. The authors find that 58% of the investigated studies show no significant relation, 27% show a positive relation, 2% show a negative relation, and 13% do not report a sample size.

Several compelling theories attempt to provide a better insight into why and how firms engage in CSR projects. The conflict resolution hypothesis stems from the stakeholder theory. It states that top managers use CSR as a device to develop and build good relationships with the firm's stakeholders. A good relationship with the employees could yield productivity increases. A good relationship with the regulatory agencies and the community could save the firm the costs associated with litigation fees, consumer boycotts, reputation damage, and regulatory fines. Excellent product quality also increases the likelihood of satisfied return customers and decreases the costs associated with product defects.

The resource-based view theory is often cited as the process through which CSR engagement could generate financial benefits for the firm [21–24]. These benefits could be achieved through the improvement of stronger and better relationships with customers [25], enhancement of future revenue growth [26], motivation and enhancement of employees' morale that could increase productivity [27], or increase in investor support [28] as well as providing "insurance-like" protection for the firm and its shareholders when an adverse event occurs [29,30]. In sum, the resource-based view posits that, by engaging in CSR, firms procure legitimacy in the public eye that they can strategically capitalize on to enhance their value.

The above two theories, conflict resolution and the resource-based viewpoint, explain CSR's value-enhancing characteristics and how firms can efficiently and effectively use them. Thus, if CSR is value-enhancing, CEOs with a golden parachute would pursue more of these investments. As a result, this increased pursuit should also enhance the value of the CEOs' golden parachute and long-term wealth.

The overinvestment hypothesis argues that top managers might overinvest in CSR projects. Managers could pursue CSR to satisfy a selfish desire to burnish their reputation as good global citizens at the shareholders' expense [31,32]. That is, managers might seek to overinvest in CSR, regardless of its effect on the shareholders' wealth, for their private gains. Consistent with the overinvestment hypothesis, some studies find a positive association between CSR and the CEO's compensation [33,34].

The strategic-choice hypothesis posits that incumbent managers strategically select CSR projects in which to invest. The motivation is to generate likability and support from environmental and social activists to reduce the manager's employment risk by decreasing the probability of future turnover [35,36]. The strategic-choice hypothesis argues that when the stakeholders' protection relies on managers' discretionary initiatives, the outcome could increase managerial opportunism. The authors of [36] assert that relations with social activists could become an effective entrenchment strategy for inefficient managers. That is, in a desperate attempt to build a good relationship with social activists, the manager could strategically choose to invest in selected CSR projects even if doing so reduces the shareholders' wealth. Such CSR projects could be in the form of donations to charitable organizations that bring positive visibility to the manager, or the funding of a costly overelaborate campaign to promote diversity among the firm's workforces.

The overinvestment and strategic choice theories both caution against the potential value-destroying characteristics of CSR engagements. Thus, CEOs with a golden parachute would likely frown on such CSR policies because they would most likely pose a risk to the value of their golden parachute and long-term wealth.

# 2.2. Executive Compensation, CSR, and Firm Performance

Several studies on related topics have explored the relationship between CSR, firm value, and various firm characteristics. They provide insights into the potential effects of factors such as ownership concentration and CEO incentives on the relationship between CSR and firm value [17,37].

Executive compensation can serve as a critical tool in aligning the manager's interest with that of the shareholders. If well designed, it could also help monitor the activities and steer the manager's decision making that affects the firm [38]. On the other hand, this monitoring tool could unintentionally restrict the manager's decision making, negatively and adversely affecting the degree of CSR commitment. With the uncertainty surrounding CSR payoffs, the decisions or agreements on allocating resources to CSR projects could prove challenging for firms [39].

Presumably, firms design fixed and shorter incentive arrangements to focus the managers' attention on projects geared towards immediate performance. Short-term performance values are usually captured using retrospective accounting numbers [40]. On the other hand, a longer-term compensation arrangement is more focused on the expectation that managers optimize their value in the long run. Thus, it should motivate managers to invest in CSR engagements [41]. Longer-term compensation arrangements are usually based on market valuation. When rightfully and efficiently implemented, CSR investments could drive the firm's value upwards long-term.

Suppose a difference of opinion between the manager and the shareholders on the CSR commitment is significantly large. In that case, this difference could infringe on the flow and success of the CSR campaign. A CSR commitment above the shareholders' willingness to commit could have negative consequences for the manager if earnings targets are not met due to the costs of the CSR investment. Thus, the manager might become reluctant to assume the risk associated with the uncertainty of the CSR payoffs. Job security and compensation risk are some of the risks that managers face.

Is the manager protected from these risks? The manager's concerns about the repercussion of assuming higher risk by investing in CSR engagements could be alleviated by a type of compensation mechanism that insulates the manager from employment risk. The provision of a golden parachute in the executive compensation contract requires that substantial benefits be given to the manager if their employment is terminated. The manager could become more willing to explore and promote CSR investments with such a condition. Or the insulation that the GP provides could embolden CEOs to reduce CSR focus, knowing that the timing of its value-enhancing attributes may not be in alignment with the CEOs' immediate personal financial interest.

Firms design long-term executive compensation schemes to focus their managers' efforts on optimizing their long-term goals and value [42]. Reference [40] finds a relationship between long-term compensation and CSR weakness. That is, long-term executive compensation, if well designed, could reduce the risk associated with CSR noncompliance. Female executives are assumed to be more risk-averse than their male counterparts. Given that the CSR cost to the firm and the benefits, if any, are not always in alignment, this could create uncertainties that a risk-averse executive may prefer to avoid. Motivated by this line of reasoning, we examine female executives' behavior towards CSR when they have a GP and when they do not have a GP.

#### 2.3. Gender and Corporate Decision-Making

Gender diversity can influence creativity in a number of different ways. Research has demonstrated that diverse teams, including gender diversity, can produce value-enhancing innovations that drive corporate breakthroughs. Gender diversity brings to the table diverse viewpoints, rich and broad experiences, and creative problem-solving methods [43–46]. However, studies have shown that women tend to be more compassionate than men [47]. These tendencies are reflected in their business and corporate decision-making as it pertains to leadership style, consensus decision-making, and risk taking [48]. Firms must be willing to take optimal levels of risk to maximize both financial and non-financial performance. Women, being more risk-averse, could adversely impact corporate decision-making opportunities that enhance firm financial performance. In business environments where shareholders' expectations are focused on value creation and management's ability to drive stock appreciation, the female tendencies of being more compassionate and risk-averse may be viewed as weaknesses that hamper firm financial performance. So, this could explain why women who rise to positions of power in male-dominated fields may feel pressure to adopt traditionally masculine behaviors in order to be seen as competent and assertive leaders.

There are several related theories that explain why women in male-dominated fields may feel pressure to adopt traditionally masculine behaviors. Self-verification theory suggests that individuals have a need to be seen and treated in ways that are consistent with their self-concept [49]. Women in male-dominated fields may adopt traditionally masculine behaviors in order to verify their sense of self as competent and capable leaders, especially in a context where they may feel that their abilities are being questioned.

Social role theory posits that individuals adopt behaviors that are consistent with the gender roles they are expected to perform [50]. In male-dominated fields, women may feel pressure to adopt traditionally masculine behaviors in order to conform to societal expectations of what a competent leader should be.

Stereotype threat theory in social psychology refers to the feeling of being in danger of confirming unfavorable preconceptions about one's group [51,52]. In the late 1990s, psychologists Claude Steele and Joshua Aronson made the initial presentation of it [53]. According to the theory, people who belong to a group that is stereotyped as possessing particular negative traits may feel anxious or apprehensive about supporting those stereotypes. Even when the person is otherwise capable of performing well, this can result in underperformance in tasks that are connected to the stereotype [53–55]. In the context of our research, if female executives are stereotyped as being less competent than their male counterparts, they may feel anxious when making investments or other related managerial decisions, which can, in turn, impair their performance.

There is also evidence in related literature that suggests that female and male executives differ systematically regarding their core values and attitudes towards risk, which differs from gender differences in the general population. Compared to their male counterparts, female executives are more security-oriented [56]. This line of thinking could shed light on the differences in behavior towards CSR when the executive is protected by a GP.

# 3. Hypothesis Development

If CSR is value-enhancing to the firm, then it should help maximize the value of the stock option component of the GP (According to Bloomberg, the largest portion of severance often comes from accelerated payments of unvested stock options and other stock units that would otherwise not be eligible for a payout—http://www.bloomberg.com/news/2013\$-\$06\$-\$06/golden-parachutes (accessed on 12 March 2023)). Executives endowed with a GP desire to maximize the value of their compensation package [39]. Therefore, the presence of a GP will moderate the association between corporate social responsibility (CSR) and firm value such that the positive relationship between CSR and firm value will be stronger for firms with a GP than those without a golden parachute.

Based on the above rationale, we propose the first hypothesis that the effect of CSR on firm value is contingent on the presence of a GP. A golden parachute is a compensation arrangement that provides executives with substantial financial benefits in the event of a change in control or a merger/acquisition. Corporate executives with a GP may be more inclined to pursue and engage in more CSR, especially if it enhances firm value, which ultimately helps maximize the value of their GPs. Therefore, we predict that the positive relationship between CSR and firm value, if it exists, will be amplified for firms with a GP compared to firms without one.

To test this hypothesis, we use a regression analysis with CSR, golden parachute, and their interaction term as independent variables and firm value as the dependent variable. The results of this analysis could indicate whether the presence of a GP moderates the association between CSR and firm value and to what extent.

# **Hypothesis 1:** Ceteris paribus, US-based firms that issue a GP to their executives will show that it amplifies the positive relationship between CSR and firm value.

Research has shown that gender-diverse teams bring a range of perspectives and experiences to the table, which can lead to better financial outcomes for the firm. Some of the value-adding attributes include more creative and effective problem-solving, better decision-making, increased innovation, better talent recruitment and retention, and enhanced firm reputation [46–48]. Despite this evidence, women face challenges navigating the corporate workplace and upward mobility ladder.

There is empirical evidence such as the social role theory [50] and self-verification theory [49] that women adopt traditionally masculine behavior to counter these stereotypes that consider them less competent. The fear or anxiety that stems from stereotype threat theory [55] may induce or pressure female executives to conform to those stereotypes, which can lead to decreased performance in areas where the stereotype is relevant. There is an expectation that corporate executives make investment decisions that positively drives firm value. In the presence of stereotype threats, in high-stakes situations, female executives may experience anxiety and reduced performance due to the fear of confirming the negative stereotype.

Drawing from these frameworks for understanding the ways in which gender characteristics and social expectations shape women's lives and experiences, we predict a difference in behavior when female executives are protected by a GP. They may feel insulated from these judgment-based social assessments and their associated consequences; ergo, they may exhibit different behavior toward CSR. It could also be that the stereotype threat prevails, consequently impacting the CSR–firm value relationship negatively.

**Hypothesis 2:** *The behavior of female corporate executives in US-based firms towards CSR will change if they are protected by a GP.* 

This study contributes to the related research on CSR and firm value by introducing the golden parachute. We posit that the presence of a golden parachute alleviates the executive's short-term concerns and incentivizes him or her to maximize the firm's performance over the long term under a firm's proactive CSR activities. Based on the arguments drawn from the previous studies, we predict that the interaction effect of both CSR and GP will be stronger than the individual effect of CSR on firm value. Accordingly, we expect the GP to amplify the positive relationship between CSR and firm value. Also, toeing the line of reasoning put forward by social role and stereotype theories that women may adopt traditionally masculine behavior to counteract the stereotypes of women as less competent than men, we predict that behavior toward CSR could change when female executives are insulated from these judgment-based social assessments.

# 4. Data and Research Model

# 4.1. Data

To examine our research question, we employed the sample period from 2007 to 2016. The CSR data were collected MSCI KLD social index. The index provides comprehensive CSR data covering about 3100 public US companies. This CSR data cover approximately 98% of the listed United States companies [32] and show the largest comprehensive CSR performance database [57]. Given their validity for multi-dimensional CSR measurement [58], KLD CSR rating data are widely used in CSR research [39,59–63].

The KLD ratings have seven categories: community, governance, diversity, employee relations, environment, human rights, and product. However, researchers consider the governance category a different construct from other CSR activities and exclude it from CSR measurements [1,64,65].

A GP is a contractual clause included in some executive compensation packages. Originally intended as a hedge against takeovers, its rise and adoption in executive compensation package contracts reflect more than a three hundred percent growth [38]. A golden parachute remains a mainstay. GP data come from the institutional shareholder services database (IRRC). GP is a dummy variable, where 1 is an indicator that the executive has adopted a golden parachute.

We employ Tobin's Q (*TOBINQ*) to measure firm value because Tobin's Q is widely accepted as a firm's long-term performance measure [63,66–69], and CSR is highly likely to pay off in the long run [70]. Moreover, management can easily manipulate accounting performance measures such as ROA by employing different accounting methods or policies.

We include control variables following prior studies to influence the CSR–firm value relationship in our model. We control for firm size (*SIZE*) [66,71,72] and sales growth rate (*GROWTH*) in that their possible effect on firms' stock price. We expect a positive relationship between *GROWTH* and firm value. We also control financial leverage (*LEVER-AGE*) [66,71]. A firm's profitability can positively influence the return on assets (ROA) in the stock price. Then, we include a firm's business risk (*RISK*) [16], *R&D\_INTENSITY* [73,74], *GOVERNANCE* [75,76], and firm age (*FIRM\_AGE*) [77,78]. Following prior studies, we also include CEOs' characteristics, such as age [79], service period [80,81], and gender [82], affecting a firm's financial performance. Lastly, we include industry effects and year effects to control for the possible influence on the relationship between CSR and firm value [71].

#### 4.2. Research Model

To investigate the moderating role of GP in the relationship between *CSR* and firm value, we use the following model (1):

$$TOBINQ_{i,t} = \alpha_0 + \alpha_1 CSR_{i,t} + \alpha_2 GP_{i,t}$$

 $+ \alpha_{3} CSR \times GP_{i,t} + \alpha_{4} SIZE_{i,t} + \alpha_{5} LEVERAGE_{i,t} + \alpha_{6}GROWTH_{i,t} + \alpha_{7} ROA_{i,t} + \alpha_{8} RISK_{i,t}$  $+ \alpha_{9} R&D_INTENSITY_{i,t} + \alpha_{10} GOVERNANCE_{i,t} + \alpha_{11} FIRM_AGE_{i,t} + \alpha_{12} CEO_AGE_{i,t}$  $+ \alpha_{13} CEO_TENURE_{i,t} + \alpha_{14} CEO_GENDER_{i,t} + \sum IND + \sum YEAR + \varepsilon_{i,t}$ (1)

where for a company *i* in period *t*, the variables are defined in the table below.

Variable	Definition
Dependent variables TOBINQ	
Variables of interest	=(market value of equity + total liabilities)/book value of total assets, at fiscal year-end in the period;
vallables of interest	annuation of accuration the community dimension analysis and time and the contract house with the cond
CSR	=summation of scores for the community, diversity, employee relations, environment, human rights, and product in KLD's ratings;
GP	=a binary variable that equals 1 if the firms have adopted golden parachutes in that year and 0 otherwise;
<b>Control variables</b>	
SIZE	=natural logarithm of total assets at the end of the fiscal year;
LEVERAGE	= total liability divided by total assets at the end of the fiscal year;
GROWTH	=net sales minus net sales at the prior year and divided by the net sales at the end of the fiscal year;
ROA	=operating income divided by total assets at the end of the fiscal year;
RISK	=standard deviation of EBIT (earnings before interests, taxes) from the past five years divided by total assets
KISK	at the end of the fiscal year;
R&D_INTENSITY	=R&D expenses divided by net sales at the end of the fiscal year;
GOVERNANCE	=net score (strengths minus concerns) in KLD governance ratings;
FIRM_AGE	=natural logarithm of (1 + the number of years since the firm's first appearance in the Compustat database);
CEO_AGE	=the age of the CEO;
CEO_TENURE	=how long the CEO has been serving in that capacity for the firm at the end of the fiscal year;
CEO_GENDER	=the gender of the CEO. A dummy variable that equals 1 if female and 0 if male;
IND	=industry indicator based on the two-digit SIC code;
YEAR	=year indicator.

# 5. Result Analysis and Endogeneity Test

# 5.1. Univariate Result

Table 1 offers a firm-year description of our sample, showing even distribution except for 2007. Our sample consists of 11,065 firm-year observations. The number of firms is spread almost evenly across the years except for 2007. In 2007, relatively more companies were omitted than in other years while merging Compustat data and KLD CSR data, which accounts for 5.62% of the total sample. Finally, the average number of firms used in our sample is 1229 (11,065/9 years). Table 2 presents the univariate results of the firms in our sample. We begin by displaying a summary statistic for the sample, shown in panel A. Then, panel B offers a stylized overview of the sample by decomposing it into two groups: firm-year observation reflecting the adoption of a GP and firm-year observation with no adoption of a golden parachute. Panel C shows the correlation and significance of each variable.

Table 1. Sample Distribution of Firm-Year Observations by Year.

Year	# of Firms	% of Sample
2007	622	5.62
2008	1099	9.93
2009	1135	10.25
2010	1133	10.24
2011	1174	10.62
2012	1183	10.69
2013	1136	10.27
2014	1195	10.80
2015	1169	10.56
2016	1219	11.01
Total	11,065	100.00

Table 2. Descriptive Statistics of Selected Variables.

Full Sample		-	Maan	Madian	011	Darr	25th Banaanti'l	754h Bana 11
Variables		n	Mean	Median	Std	. Dev.	25th Percentile	75th Percentil
Dependent Variables	TOBINQ	11.065	1.842	1.481	1	154	1 1 2 9	2.117
Variable of Interest	TOBINQ	11,065	1.842	1.481	1	154	1.128	2.117
variable of interest	CSR	11,065	0.793	0.000	2	628	-1.000	2.000
	GP	11,065	0.742	1.000		438	0.000	1.000
Control Variables								
	SIZE	11,065	8.140	8.054	1	571	6.959	9.230
	LEVERAGE	11,065	0.559	0.556	0	232	0.405	0.709
	GROWTH	11,065	0.063	0.048	0	220	-0.024	0.130
	ROA	11,065	0.090	0.081	0	088	0.042	0.130
	RISK	11,065	0.032	0.021	0	053	0.010	0.037
	R&D_INTENSITY	11,065	0.030	0.000	0	111	0.000	0.022
	GOVERNANCE	11,065	-0.205	0.000	0	675	-1.000	0.000
	FIRM_AGE	11,065	3.207	3.219	0	626	2.833	3.807
	CEO_AGE	11,065	57.488	57.000	7	056	53.000	62.000
	CEO_TENURE	11,065	11.043	8.000	9	065	4.000	15.000
	CEO_GENDER	11,065	0.039	0.000		195	0.000	0.000
Descriptive Statistics Measured by Adoptic	on of Golden Parachute	5						
							Differe	nce Tests:
	Ado	ption of Golden Parach	utes	Non-A	Adoption of Golden Pa	rachutes		alue
Variables	n	Mean	Median	n	Mean	Median	t-test	Wilcoxon Test
TOBINQ	8211	1.817	1.468	2854	1.912	1.523	<0.001	0.0789
CSR	8211	0.855	0.000	2854	0.615	0.000	< 0.001	< 0.001
SIZE	8211	8.162	8.089	2854	8.079	7.944	0.022	0.001
LEVERAGE	8211	0.569	0.568	2854	0.530	0.523	< 0.001	< 0.001
GROWTH	8211	0.061	0.046	2854	0.069	0.055	0.079	0.005
ROA	8211	0.087	0.078	2854	0.099	0.091	< 0.001	< 0.001
RISK	8211	0.031	0.020	2854	0.034	0.024	0.004	< 0.001
R&D_INTENSITY	8211	0.031	0.000	2854	0.026	0.000	0.007	< 0.001
GOVERNANCE	8211	-0.180	0.000	2854	-0.279	0.000	< 0.001	< 0.001
FIRM AGE	8211	3.218	3.219	2854	3.177	3.219	0.002	0.002
CEO_AGE	8211	57.368	57.000	2854	57.836	57.000	0.005	0.051
CEO_TENURE	8211	10.231	8.000	2854	13.378	10.000	< 0.001	< 0.001
CEO_GENDER	8211	0.041	0.000	2854	0.034	0.000	0.071	0.084
Correlation Coefficients among Selective Va	ariables (N = 11,065)							
-		1	2	3	4	5	6	7
	TOBINO	1.000						
2	CSR	0.076 **	1.000					
3	GP	-0.036 **	0.040 **	1.000				
-	SIZE	-0.250 **	0.376 **	0.023 *	1.000			
	LEVERAGE	-0.182 **	0.121 **	0.073 **	0.515 **	1.000		
	GROWTH	0.161 **	-0.043 **	-0.017	-0.053 **	-0.079 **	1.000	
	ROA	0.550 **	0.058 **	-0.059 **	-0.149 **	-0.192 **	0.200 **	1.000
	RISK	0.177 **	-0.067 **	-0.028 **	-0.240 **	-0.179 **	-0.011	0.035 **
	R&D INTENSITY	0.154 **	-0.007	0.022 *	-0.202 **	-0.186 **	0.042 **	-0.158 **
0	GOVERNANCE	0.024 *	0.151 **	0.064 **	-0.027 **	0.014	-0.013	-0.021 *
1	FIRM_AGE	-0.122 **	0.178 **	0.029 **	0.311 **	0.123 **	-0.136 **	-0.008
12	CEO_AGE	-0.078 **	-0.009	-0.029 **	0.080 **	0.037 **	-0.035 **	-0.023 *
12	CEO_AGE CEO TENURE	0.032 **	-0.100 **	-0.152 **	-0.127 **	-0.124 **	0.024*	0.006
14	CEO_TENURE CEO_GENDER	-0.003	0.086 **	0.016	0.014	0.033 **	-0.028 **	0.008
17	CLO_GLINDER	8	9	10				
,	DIGH		У	10	11	12	13	14
3	RISK R&D INTENSITY	1.000	1.000					
		0.132 ** -0.025 **	1.000	1.000				
0	GOVERNANCE		-0.017	1.000	1 000			
1	FIRM_AGE	-0.149 **	-0.085 **	0.041 **	1.000	1 000		
2	CEO_AGE	-0.020*	-0.054 **	0.016	0.140 **	1.000	1 000	
3	CEO_TENURE	-0.017	0.003	0.037 **	0.030 **	0.499 **	1.000	1.000
4	CEO_GENDER	-0.01	-0.028 **	0.004	0.013	-0.052 **	-0.077 **	1.000

Pearson correlation coefficients are shown below the diagonal. \* and \*\* represent statistical significance at the 5% and 1% levels, respectively, on a two-tailed test. Variables are defined in Appendix A. Variable definitions: TOBINQ = (market value of equity + total liabilities)/book value of total assets, at fiscal year-end in period; CSR = summation of scores for community, diversity, employee relations, environment, human rights, and product in KLD's ratings;*GP*= a binary variable that equals 1 if the firms have adopted golden parachutes in that year and 0 otherwise;*SIZE*= natural logarithm of total assets at the end of the fiscal year;*LEVERAGE*= total liability divided by total assets at the end of the fiscal year;*ROA*= operating income divided by total assets at the end of the fiscal year;*RiSK*= standard deviation of EBIT (earnings before interests, taxes) from the past five years divided by total assets at the end of the fiscal year;*RCD\_INTENSITY*= R&D expenses divided by ne sales at the end of the fiscal year;*RCD\_INTENSITY*= R&D expenses divided by ne sales at the end of the fiscal year;*RCD\_INTENSITY*= R&D expenses divided by ne sales at the end of the fiscal year;*RCD\_INTENSITY*= R&D expenses divided by ne sales at the end of the fiscal year;*COVERNANCE*= net score (strengths minus concerns) in KLD governance ratings;*FIRM\_AGE*= natural logarithm of (1 + the number of years since the firm's first appearance in the Compustat database);*CEO\_AGE*= the age of the CEO;*CEO\_TENURE*= how long the CEO has been serving in that capacity for the firm at the end of the fiscal year;*CEO\_GENDER*= the gender of the CEO.

In Table 2, Full Sample, we find that the mean of CSR's variable is slightly positive (0.793), suggesting that the sample contains firms with more CSR strengths than CSR weaknesses. The mean of the GP variable (0.742) suggests that more than seventy-four percent of the firm-year observations have the GP clause for their executives.

In Table 2, Descriptive Statistics Measured by Adoption of Golden Parachutes, we offer a stylized summary and a univariate result comparing the differences between variables for firms that have adopted GPs and non-GP firms. We test the two categories using a parametric test and a non-parametric test. The evidence shows that the mean of *TOBINQ* is different for the GP and non-GP firms (1.817 vs. 1.912, p < 0.001). That is, firms without a GP have a slightly higher value than GP firms. Interestingly, the mean and median figures show that non-GP firms have a larger *TOBINQ* than GP firms. Given that we are simply comparing the mean and median, we should reserve any conclusions until we see the results of the regression analysis with control variables. The contrast in CSR and its breakdown shows a gleaning behavior for GP firms and non-GP firms. GP firms pursue CSR initiatives more than their counterparts.

In Table 2, Correlation Coefficients, the correlation matrix shows that all variables have a significant relationship with the explained proxy for firm value, TOBINQ, except the control variable, CEO gender, is not significant. As we expected, the firm value is positively correlated with *CSR* (0.076, *p* < 0.01), *GROWTH* (0.161, *p* < 0.01), *ROA* (0.550, p < 0.01), RISK (0.177, p < 0.01), R&D\_INTENSITY (0.154, p < 0.01), GOVERNANCE (0.024, p < 0.05), and CEO\_TENURE (0.032, p < 0.01), whereas the firm value is negatively correlated with *GP* (-0.036, *p* < 0.01), *LEVERAGE* (-0.182, *p* < 0.01), *FIRM\_AGE* (-0.122, *p* < 0.01), and CEO\_AGE (-0.078, p < 0.01). The negative correlation between GP and TOBINQ may allude to why some research findings suggest that the adoption of GP serves as an entrenchment clause [83,84]. Here, the correlation suggests that the GP's adoption reduces a firm's investment opportunities without considering the effects of control variables in the regression model. Moreover, we check whether the correlation coefficients are higher than 0.5, which may cause multicollinearity. We find the correlation coefficient between SIZE and LEVERAGE is 0.515. However, these two independent variables are necessary control variables shown in prior studies [1,64,65]. Further, we measure the variation inflation factor (VIF) in our regression and find no material multicollinearity issue in our regression models.

# 5.2. Regression Results

We display the regression results. We show the beta coefficient; thus, we can infer the key variables' significant impact on the dependent variable. In Table 3, Model 1, we find that *CSR* is positively associated with *TOBINQ* (0.090, p < 0.01). Model 2 shows a negative relationship between *GP* and *TOBINQ* (-0.044, p < 0.01), which is consistent with prior studies [83]. *GP* is a dummy variable that takes the value of 1 if the firm's executive has a *GP* and 0 if not. More importantly, in Model 4, we explore the interaction of *GP* with *CSR* (*CSRXGP*). The coefficient of the interaction variable (*CSRXGP*) is 0.027 (p < 0.01), implying that *GP* strengthens the positive relationship between *CSR* and firm value, *TOBINQ*. *GP* plays an important moderating role in the relationship between CSR and firm value. This result lends credence to our argument that the presence of a golden parachute alleviates the executive's short-term concerns and incentivizes him or her to maximize the firm's performance over the long term. Accordingly, this study supports our Hypothesis 1 that the *GP* amplifies the positive relationship between CSR and firm value.

Further, we divide the sample into two groups: adoption of GP and non-adoption of GP. By dividing our sample, we can measure the extent to which CSR activities affect the firm value in each different group to see the level of impact of the adoption of GP on the relationship between *CSR* and *TOBINQ*. In Model 5 and Model 6, we find that the coefficient and t value of the GP adoption group are much greater than those of the GP non-adoption group (0.108, t = 10.01 vs. 0.029, t = 1.85, respectively). This result supports the interaction effect of CSR and GP on the firm value.

In Table 4, we report the results of the regression when six individual components of *CSR* are separately regressed on *TOBINQ*. Among the six interaction variables, the coefficients (*p*-value) of *COMMUNITYXGP* (Model 2), *HUMAN\_RIGHTXGP* (Model 10), and *PRODUCTXGP* (Model 12) are positive and significant (0.017, *p* < 0.10; 0.027, *p* < 0.05; 0.057, *p* < 0.01, respectively). These results suggest that a firm's GP adoption plays a more critical moderating role in the community, human rights promotion, and product quality activities enhancing firm value. This result confirms the robustness of our findings in Table 3.

						Adoption of GP	Non-Adoption of G		
		Dependent Variable: TOBINQ							
		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6		
Independent variables	Predicted Sign	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)		
CSR	+	0.090 ***		0.089 ***	0.067 ***	0.108 ***	0.029 *		
GP	_	(9.92)	-0.044 ***	(9.82) -0.043 ***	(5.48) -0.047 ***	(10.01)	(1.85)		
$CSR \times GP$	+		(-5.58)	(-5.39)	(-5.73) 0.027 *** (2.58)				
SIZE	?	-0.141 ***	-0.099 ***	-0.144 ***	-0.143 ***	-0.178 ***	-0.064 **		
LEVERAGE	+	(-8.37) 0.069 ***	(-6.61) 0.068 ***	(-8.55) 0.073 ***	(-8.52) 0.072 ***	(-9.02) 0.110 ***	(-2.49) -0.052 **		
GROWTH	+	(2.76) 0.037 ***	(2.68) 0.032 ***	(2.91) 0.037 ***	(2.87) 0.037 ***	(3.58) 0.040 **	(-2.29) 0.020		
ROA	+	(3.09) 0.530 *** (12.50)	(2.65) 0.536 ***	(3.08) 0.529 ***	(3.09) 0.530 ***	(2.48) 0.508 ***	(1.51) 0.603 *** (1< 01)		
RISK	+	(12.50) 0.110 *** (4.21)	(12.55) 0.113 ***	(12.49) 0.109 *** (4.27)	(12.48) 0.109 *** (4.27)	(9.76) 0.127 *** (4.21)	(16.81) 0.041 (1.10)		
R&D_INTENSITY	+	(4.31) 0.172 *** (4.20)	(4.34) 0.176 *** (4.23)	(4.27) 0.172 *** (4.22)	(4.27) 0.172 *** (4.23)	(4.21) 0.166 *** (3.72)	(1.16) 0.220 *** (4.28)		
GOVERNANCE	+	(-0.029 ***) (-4.32)	(4.23) -0.016 ** (-2.38)	(4.22) -0.029 *** (-4.31)	(4.23) -0.030 *** (-4.39)	(0.72) -0.032 *** (-3.83)	(-0.031 ** (-2.56))		
FIRM_AGE	_	(-0.047 ***) (-5.62)	(-0.043 ***) (-5.08)	(-0.047 ***) (-5.65)	$(-0.047^{***})$ (-5.64)	-0.036 *** (-3.69)	(-0.059 *** (-3.41))		
CEO_AGE	_	-0.067 *** (-6.95)	-0.066 *** (-6.87)	-0.066 *** (-6.85)	-0.066 *** (-6.86)	-0.052 *** (-5.04)	-0.084 *** (-4.31)		
CEO_TENURE	+	0.068 *** (7.20)	0.056 *** (5.99)	0.061 *** (6.54)	0.061 *** (6.49)	0.050 *** (4.73)	0.056 *** (3.13)		
CEO_GENDER	_	-0.020 *** (-3.34)	-0.013 ** (-2.25)	-0.019 *** (-3.29)	(-0.019 *** (-3.24)	$-0.023^{***}$ (-3.32)	0.003 (0.23)		
Industry dummies Year dummies		Included Included	Included Included	Included Included	Included Included	Included Included	Included Included		
Mean of VIF Adj. R <sup>2</sup>		1.25 0.47	1.21 0.47	1.24 0.48	1.53 0.48	1.25 0.46	1.25 0.55		
F		71.97 *** 11,065	72.21 *** 11,065	71.13 *** 11,065	70.43 *** 11,065	52.73 *** 8211	43.34 *** 2854		

\*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively, on a two-tailed test. All of the test statistics and significance levels are calculated based on robust standard errors adjusted for heteroskedasticity at both the firm and year levels.

# 5.3. Regression Results: Does Gender Play a Role?

Furthermore, in Table 5, we divide the data into female and male groups to test *Hypothesis 2: The behavior of female corporate executives in US-based firms towards CSR will change if they are protected by a GP*. The results show that for firms with female CEOs who are protected by GPs, the relationship between CSR and firm value is negative but not statistically significant (-0.064, t = -1.34). However, for the male CEO group, the strong positive moderating power of the GP is sustained (0.031, *p* < 0.01), supporting *Hypothesis 2*. Related theories could explain this contrasting finding on why women in male-dominated fields may feel pressure to adopt traditionally masculine behaviors, especially on their rise to the top of the corporate ladder.

Self-verification theory and social role theory both suggest that women in maledominated fields may feel pressure to adopt traditionally masculine behaviors because of societal expectations, the threat of being judged based on gender stereotypes, and a desire to maintain their sense of self. Stereotype theory can lead to underperformance in tasks that are related to the stereotype, even when the individual is otherwise capable of performing well. The effects of stereotype threat can be significant and can contribute to persistent disparities in achievement and performance between different social groups.

							Dependent Va	riable: TOBIN	Q				
		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6	Model 7	Model 8	Model 9	Model 10	Model 11	Model 12
Independent Variables	Predicted Sign	Coeff. (t-stat)	Coeff. (t-stat)										
COMMUNITY	+	0.033 ***	0.019*										
COMMUNITYXGP	+	(5.35)	(1.83) 0.017 * (1.80)										
DIVERSITY	+			0.047 *** (5.49)	0.036 *** (3.14)								
DIVERSITYXGP	+			(0.17)	0.014 (1.40)								
EMPLOYEE	+					0.062 *** (6.84)	0.056 *** (4.32)						
EMPLOYEEXGP	+					(0.04)	0.007						
ENVIRON	+						(0.56)	0.053 ***	0.062 ***				
ENVIRONXGP	+							(6.95)	(5.22) -0.015				
HUMAN_RIGHT	+								(-1.47)	0.019 **	-0.004		
HUMAN_RIGHTXGP	<b>*</b> +									(1.96)	(-0.37) 0.027 ** (2.14)		
PRODUCT	+										()	0.024 *** (3.09)	-0.021 ** (-2.07)
PRODUCTXGP	+											(3.09)	0.057 ***
GP			-0.046 ***		-0.046 ***		-0.044 ***		-0.038 ***		-0.043 ***		(5.23) -0.039 ***
Control variables Industry dummies Year dummies Adj. R <sup>2</sup> F n		Included Included Included 0.47 72.42 *** 11,065	(-5.62) Included Included 0.47 70.82 *** 11,065	Included Included Included 0.47 72.41 *** 11,065	(-5.72) Included Included 0.47 70.82 *** 11,065	Included Included Included 0.47 72.22 *** 11,065	(-5.61) Included Included 0.47 70.69 *** 11,065	Included Included Included 0.47 75.34 *** 11,065	(-4.45) Included Included 0.47 73.68 *** 11,065	Included Included Included 0.47 72.45 *** 11,065	(-5.46) Included Included 0.47 70.64 *** 11,065	Included Included Included 0.47 72.42 *** 11,065	(-4.73) Included Included 0.47 70.29 *** 11,065

Table 4. Regressions of *TOBINQ* on Individual Components of *CSR and GP* (Golden Parachute).

\*\* and \*\*\* indicate statistical significance at the 5% and 1% levels, respectively, on a two-tailed test. All of the test statistics and significance levels are calculated based on robust standard errors adjusted for heteroskedasticity at both the firm and year levels. Variable definition: *COMMUNITY* = a firm's community activities such as charitable giving and innovative giving; *DIVERSITY* = a firm's diversity activities such as CEO gender and the promotion of women and minorities: *EMPLOYEE* = a firm's employee relations such as union relations and employee involvement; *ENVIRON* = a firm's environmental activities such as pollution prevention and clean energy; *HUMAN\_RIGHT* = a firm's human rights activities; *PRODUCT* = a firm's product quality activities. All variables are measured by strengths minus concerns. Other variables are defined in Appendix A.

				Dependent Va	riable: TOBIN(	2	
		I	Female CEO Gro	oup		Male CEO Gro	up
		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
Independent Variables	Predicted Sign	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)
CSR	+	0.105 *** (3.33)		0.157 *** (2.79)	0.089 *** (9.61)		0.063 *** (5.11)
GP	_		-0.054 * (-1.69)	-0.006 (-0.14)		-0.042 *** (-5.21)	-0.046 *** (-5.50)
$CSR \times GP$	+		. ,	-0.064 (-1.34)		. ,	0.031 *** (2.81)
SIZE	?	-0.111 (-1.58)	-0.056 (-0.92)	-0.126 * (-1.71)	-0.140 *** (-8.07)	-0.097 *** (-6.34)	-0.142 *** (-8.21)
LEVERAGE	+	0.170 *** (3.60)	0.180 *** (3.70)	0.171 *** (3.60)	0.065 ** (2.48)	0.063 ** (2.38)	0.068 *** (2.59)
GROWTH	+	0.075 (1.41)	0.064 (1.30)	0.072 (1.34)	0.037 *** (2.99)	0.032 ** (2.56)	0.037 *** (2.99)
ROA	+	0.596 *** (11.12)	0.598 <sup>***</sup> (11.29)	0.592 *** (11.05)	0.526 *** (12.06)	0.532 *** (12.11)	0.525 *** (12.04)
RISK	+	0.145 ** (2.42)	0.139 ** (2.32)	0.143 ** (2.41)	0.109 *** (4.23)	0.112 *** (4.27)	0.108 *** (4.19)
R&D_INTENSITY	+	0.165 *** (2.85)	0.171 *** (2.72)	0.163 *** (2.78)	0.171 *** (4.14)	0.175 *** (4.17)	0.171 *** (4.16)
GOVERNANCE	+	-0.006 (-0.24)	0.016 (0.62)	-0.007 (-0.29)	-0.031 *** (-4.46)	-0.018 *** (-2.61)	-0.032 *** (-4.54)
FIRM_AGE	-	-0.042 (-1.09)	-0.048 (-1.21)	-0.044 (-1.13)	-0.048 *** (-5.66)	-0.044 *** (-5.10)	-0.048 *** (-5.65)
CEO_AGE	-	-0.000 ( $-0.01$ )	0.006 (0.14)	-0.004 (-0.09)	$-0.069^{***}$ (-7.01)	$-0.069^{***}$ (-6.94)	$-0.068^{***}$ (-6.91)
CEO_TENURE	+	0.089 ** (2.40)	0.093 ** (2.39)	0.087 ** (2.34)	0.071 *** (7.37)	0.059 *** (6.18)	0.064 *** (6.70)
Industry dummies		Included	Included	Included	Included	Included	Included
Year dummies		Included	Included	Included	Included	Included	Included
Adj. R <sup>2</sup>		0.65	0.65	0.65	0.47	0.47	0.47
, F		15.50 ***	15.13 ***	14.99 ***	71.10 ***	71.30 ***	69.32 ***
n		436	436	436	10,629	10,629	10,629

Table 5. Regressions of TOBINQ on CSR and GP (Golden Parachute) in Female and Male Groups.

\*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively, on a two-tailed test. All of the test statistics and significance levels are calculated based on robust standard errors adjusted for heteroskedasticity at year levels.

In Table 6, we employ the propensity-score-matching model to mitigate the endogeneity issue. The propensity-score-matched samples seem effective in obtaining a balanced set of samples [85]. We impose a caliper distance of 0.001 using *SIZE* and *ROA* variables to calculate the propensity scores to obtain a sample of 1599 for the male CEO group out of 10,629 firm-year observations to match with a sample of 436 for the female CEO group. In Table 6, Model 1, we show the result using the full sample of 2035 after applying the PSM methodology. The coefficient of CSR × GP is statistically significant (0.043, p < 0.10), supporting Hypothesis 2. We also report in Models 2 and 3 that for firms with female executives protected by the GP, the relation between CSR and firm value is negative but not statistically significant (-0.064, t=-1.34). However, for the male executives' group, the strong positive moderating power of the GP is still maintained (0.058, p < 0.05), supporting *Hypothesis* 2.

Dependent Variable: TOBINQ						
		Full Sample	Female Group	Male Group		
Independent Predicted Variables Sign		Model 1	Model 2	Model 3		
		Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)		
CSR	+	0.028	0.157 ***	0.001		
GP	_	(1.13) -0.083 ***	(2.79) -0.006	(0.04) -0.079 ***		
CSR  imes GP	+	(-4.10) 0.043 *	(-0.14) -0.064	(-3.52) 0.058 **		
CEO_GENDER	_	(1.81) -0.034 **	(-1.34)	(2.05)		
SIZE	?	(-2.33) -0.099 ***	-0.126 *	-0.087 ***		
LEVERAGE	+	(-3.78) 0.069 ***	(-1.71) 0.171 ***	(-3.15) 0.047*		
GROWTH	+	(2.87) 0.020	(3.60) 0.072	(1.71) 0.020		
ROA	+	(0.98) 0.652 ***	(1.34) 0.592 ***	(0.88) 0.662 ***		
RISK	+	(21.85) -0.000	(11.05) 0.143 **	(18.89) -0.020		
<i>R&amp;D_INTENSITY</i>	+	(-0.01) 0.161 ***	(2.41) 0.163 ***	(-0.53) 0.169 ***		
GOVERNANCE	+	(5.37) -0.013	(2.78) -0.007	(5.94) -0.023		
FIRM_AGE	_	(-0.96) -0.046 **	(-0.29) -0.044	(-1.56) -0.047 **		
CEO_AGE	_	(-2.32) -0.069 ***	(-1.13) -0.004	(-2.05) -0.082 ***		
_ CEO_TENURE	+	(-2.96) 0.067 ***	(-0.09) 0.087 **	(-3.11) 0.081 ***		
Industry dummies	·	(3.26) Included	(2.34) Included	(3.52) Included		
Year dummies Adj. R <sup>2</sup>		Included 0.59	Included 0.65	Included 0.59		
F n		34.79 *** 2035	14.99 *** 436	28.95 *** 1599		

**Table 6.** Regressions of *TOBINQ* on *CSR and GP* in Full, Female, and Male Group Propensity Score Matching Method.

\*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively, on a two-tailed test. All of the test statistics and significance levels are calculated based on robust standard errors adjusted for heteroskedasticity at year levels.

#### 5.4. Endogeneity Tests

In the CSR-firm value study, we need to consider the endogeneity problem to explain cause-and-effect relationships because of reverse causality issues, omitted variables, and measurement errors. For example, we may expect highly profitable firms to allocate more capital and other resources toward CSR activities. It could also be that proactive CSR firms are more successful in maximizing firm value. Therefore, it is unclear if CSR is the cause or the effect. To address this endogeneity issue, we conduct a two-stage least-square regression (2SLS) [3,4,86,87] using an instrumental variable (IV) for CSR [88–90].

The proper IV needs to be highly correlated with CSR but less correlated to the firm value variable (*TOBINQ*). We use the two-digit SIC industry average CSR score as IV following a prior study [88]. The industry average CSR score can be positively correlated with a firm's CSR score yet does not influence its value. The minimum eigenvalue statistic or F statistic is 1735.65, which is much higher than 10, implying that our IV seems to be a

proper instrument. The correlation coefficient between *CSR* and IV is 0.404 (p < 0.01), but the correlation coefficient between *TOBINQ* and IV is 0.102 (p < 0.01). Thus, this correlation supports the validity of our IV.

Table 7 shows the results of the 2SLS regression. In the first stage, in Model 1, we predict the value of *CSR* using IV of *CSR\_INDUSTRY\_AVG* and label that predicted value of *CSR* as *CSR\_HAT* (instrumented). Then, in the second stage of Model 2, we use *CSR\_HAT* to explore the relationship between *CSR\_HAT* and firm value. We find that the coefficients of *CSR\_INDUSTRY\_AVG* and *CSR\_HAT* variables are significantly positive (0.365, p < 0.01; 0.075, p < 0.01, respectively).

Instrument Variable: CSR_INDUSTRY_AVG						
	Model 1	Model 2				
Independent variables	Coeff. (t-stat)	Coeff. (t-stat)				
	First Stage	Second Stage				
	Dep. variable	Dep. variable				
	CSR	TOBINQ				
CSR_INDUSTRY_AVG	0.365 *** (33.31)					
CSR_HAT (instrumented)	()	0.075 *** (4.81)				
SIZE	0.411 *** (33.99)	$-0.164^{***}$ (-7.68)				
LEVERAGE	-0.087 *** (-8.50)	0.059 *** (2.62)				
GROWTH	-0.055 *** (-6.25)	0.041 *** (3.25)				
ROA	0.114 *** (10.45)	0.551 *** (13.39)				
RISK	0.033 *** (4.75)	0.110 *** (4.66)				
<i>R&amp;D_INTENSITY</i>	0.055 *** (3.54)	0.196 *** (4.32)				
GOVERNANCE	0.118 *** (10.16)	-0.039 *** (-4.90)				
FIRM_AGE	0.050 *** (6.07)	-0.061 *** (-7.45)				
CEO_AGE	-0.021 ** (-2.43)	-0.073 *** (-7.39)				
CEO_TENURE	-0.046 *** (-5.35)	0.065 *** (7.20)				
CEO_GENDER	0.068 *** (7.80)	-0.019 *** (-3.02)				
Year dummies	Included	Included				
Mean of VIF Adj. R <sup>2</sup>	1.92 0.32	2.15 0.43				
F	189.93 ***	80.24 ***				
n ** indicate statistical significance at the 5%	11,065	11,065				

Table 7. 2 SLS Regressions of *TOBINQ* on *CSR* Using Instrument Variable.

<sup>\*\*</sup> indicate statistical significance at the 5% levels, \*\*\* indicate statistical significance at the 1% levels, respectively, on a two-tailed test. All of the test statistics and significance levels are calculated based on robust standard errors adjusted for heteroskedasticity at year levels. Variable definition: *CSR\_INDUSTRY\_AVG* = an industry average score of CSR score based on the same SIC two-digit code. *CSR\_HAT* (instrumented) = a predicted value of CSR score, which is a dependent variable using independent variables including an instrument variable, *CSR\_INDUSTRY\_AVG*. Other variables are shown in Appendix A.

# 5.5. Alternative Measurement: CSR Dummy Variable

Table 8 shows the results of the regression analyses using the *CSR\_DUMMY* variable to test if the results of Table 3 are supported when we use a CSR score dummy variable instead of CSR scores itself. *CSR\_DUMMY* variable takes the value of 1 if a firm's CSR score is greater than the median number of CSR scores and 0 otherwise. In Model 1, we report that the coefficient (0.077, p < 0.01) of *CSR\_DUMMY* is positively and significantly related to *TOBINQ* as expected. Moreover, the coefficient of the interaction variable, *CSR\_DUMMY* × *GP*, is positively and significantly associated with *TOBINQ* (0.029, p = 0.059), implying that GP plays a positive moderating role in the positive relationship between *CSR* and *TOBINQ*. Further, we examine whether the adoption of GP can affect the relationship between *CSR\_DUMMY* and *TOBINQ*. To perform this test, we divided the sample into two groups: adoption of GP and non-adoption of GP. In Models 5 and 6, we report that the positive relationship between *CSR\_DUMMY* and *TOBINQ* in the firm group of adopting GP is much greater than that in the group of not adopting GP (0.095, p < 0.01 versus 0.022, t = 1.45 (not significant)). Thus, we conclude that the results of Table 3 are fully supported by the results of Table 8.

Table 8. Regressions of TOBINQ on CSR\_DUMMY and GP (Golden Parachute).

						Adoption of GP	Non-Adoption of G		
			Dependent Variable: TOBINQ						
		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6		
Independent variables	Predicted Sign	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)	Coeff. (t-stat)		
CSR_DUMMY	+	0.077 ***		0.078 ***	0.055 ***	0.095 ***	0.022		
		(9.00)		(9.16)	(3.74)	(9.35)	(1.45)		
GP	_		-0.044 ***	-0.046 ***	-0.057 ***				
			(-5.58)	(-5.83)	(-5.65)				
$CSR_DUMMY \times GP$	+				0.029 *				
CITE	2	-0.128 ***	-0.099 ***	0 100 ***	(1.89)				
SIZE	?			-0.132 ***	-0.131 ***	-0.165 ***	-0.058 **		
LEVERAGE		(-8.00) 0.066 ***	(-6.61) 0.068 ***	(-8.25) 0.070 ***	(-8.19) 0.070 ***	(-8.69) 0.108 ***	(-2.38) -0.054 **		
LEVEKAGE	+	(2.63)	(2.68)	(2.80)	(2.78)				
GROWTH		(2.63) 0.036 ***	0.032 ***	(2.80) 0.036 ***	(2.78) 0.036 ***	(3.54) 0.039 **	(-2.41) 0.020		
GKOWIII	+	(3.01)	(2.65)	(3.00)	(3.01)	(2.41)	(1.47)		
ROA	+	0.532 ***	0.536 ***	0.531 ***	0.531 ***	0.509 ***	0.604 ***		
KOM	т	(12.50)	(12.55)	(12.49)	(12.49)	(9.75)	(16.91)		
RISK	+	0.111 ***	0.113 ***	0.110 ***	0.110 ***	0.127 ***	0.042		
hibh	·	(4.34)	(4.34)	(4.30)	(4.29)	(4.21)	(1.18)		
R&D_INTENSITY	+	0.173 ***	0.176 ***	0.173 ***	0.173 ***	0.167 ***	0.220 ***		
		(4.22)	(4.23)	(4.24)	(4.24)	(3.74)	(4.27)		
GOVERNANCE	+	-0.022 ***	-0.016 **	-0.022 ***	-0.022 ***	-0.022 ***	-0.028 **		
		(-3.27)	(-2.38)	(-3.30)	(-3.36)	(-2.73)	(-2.38)		
FIRM_AGE	_	-0.044 ***	-0.043 ***	-0.045 ***	-0.045 ***	-0.032 ***	-0.058 ***		
_		(-5.31)	(-5.08)	(-5.34)	(-5.32)	(-3.34)	(-3.38)		
CEO_AGE	_	-0.068 ***	-0.066 ***	-0.066 ***	-0.066 ***	-0.053 ***	-0.083 ***		
		(-6.98)	(-6.87)	(-6.88)	(-6.88)	(-5.12)	(-4.27)		
CEO_TENURE	+	0.066 ***	0.056 ***	0.059 ***	0.058 ***	0.046 ***	0.055 ***		
		(6.99)	(5.99)	(6.28)	(6.21)	(4.44)	(3.07)		
CEO_GENDER	+	-0.020 ***	-0.013 **	-0.019 ***	-0.019 ***	-0.024 ***	0.004		
		(-3.34)	(-2.25)	(-3.32)	(-3.28)	(-3.44)	(0.30)		
Industry dummies		Included	Included	Included	Included	Included	Included		
Year dummies		Included	Included	Included	Included	Included	Included		
Adj. R <sup>2</sup>		0.47	0.47	0.47	0.47	0.46	0.55		
F		71.23 ***	72.21 ***	70.34 ***	69.61 ***	53.11 ***	43.32 ***		
n		11,065	11,065	11,065	11,065	8211	2854		

\*\*, and \*\*\* indicate statistical significance at the 5%, and 1% levels, respectively, on a two-tailed test. All of the test statistics and significance levels are calculated based on robust standard errors adjusted for heteroskedasticity at year levels. Variable definition:  $CSR_DUMMY$  = an indicator variable that takes a value of 1 if the CSR score is greater than the median number of CSR score, and 0 otherwise.  $CSR_DUMMY \times GP$  = an interaction variable measured as  $CSR_DUMMY$  times GP. Other variables are shown in Appendix A.

# 6. Discussion

Does the GP moderate the association between CSR and firm value for US-based firms? Does the gender of the executive make any difference in these associations? This study empirically examines these questions. Our findings shed much-needed light and contribute to the CSR-firm value debate, which has produced mixed results [1,4,5,7–9].

The results of our findings demonstrate the important role that gender plays in the GP adoption policy when assessing a US-based firm's CSR investment. This study presents novel findings that contribute to CSR literature; specifically, these findings are related to the alignment of CEOs' interest (GP) and the society's campaign for CSR in the US-based environment. We show how firms' long-term value, CEOs' interests, and CSR practices can be optimally achieved. A CEO with a GP will pursue and engage in value-enhancing CSR that benefits both shareholders and the society from which they extract profit. We also provide empirical evidence suggesting that in contrast to the findings for the general population, women executives endowed with a GP do not perform well on the CSR-firm value-adding assessment spectrum. The implication of this suggests that corporate CSR strategies should be adjusted based on the CEO's gender. Society benefits from CSR.

Our findings on the CSR–firm value associations support the conflict resolution and resources-based view theories. The resource base view and conflict resolution theories of CSR highlight the importance of CSR activities as a strategic tool for firms to create and sustain competitive advantages. From these theories' perspectives, CSR activities can be seen as a way to develop and leverage strategic resources and capabilities. These theories emphasize the importance of businesses as key actors in society and call for a collaborative approach to addressing social and environmental issues. For example, CSR activities such as environmental sustainability programs or philanthropic initiatives can help firms build positive relationships with stakeholders, enhance their reputation and brand image, and foster innovation and learning within the organization. Businesses can build stronger relationships and create shared value for themselves and their communities by engaging with their stakeholders and working together to resolve conflicts [20,22,91,92].

Investors and society largely frown upon firms issuing golden parachutes to CEOs. The author of [93] finds that Fortune 500 firms that adopted a golden parachute in the 1980s recorded negative stock returns. The authors of [83] also recorded negative abnormal stock returns due to firms adopting a golden parachute. Thus, investors and society seem to frown upon firms adopting a golden parachute. Nonetheless, US-based firms issue golden parachutes to attract the best talent for managerial positions and stay competitive. Despite the seemingly ever-present controversies surrounding golden parachutes, firms' issuance of golden parachutes is a practice that is here to stay [18]. Therefore, finding ways for these interests to coexist benefits the firm, the CEO, and society. Our research presents strategic alliances that benefit the interests of these groups: society, firms, and corporate executives.

Stereotype threat theory describes how individuals' performance in a task can be adversely affected by the awareness or anticipation of negative stereotypes about their group. Female executives may face a situation where their behavior or performance is likely to be evaluated based on a negative stereotype about their gender. Being consciously aware of these negative stereotypes, female executives may experience anxiety and self-doubt, leading to underperformance in the task at hand (e.g., strategically and carefully identifying value-enhancing CSR projects).

Firms should strive to create a more supportive, inclusive, and equitable environment. Measures should be put in place that actively facilitate a corporate culture that helps female employers in leadership roles to reframe their thinking about these stereotypes and their own abilities. Interventions such as affirming positive aspects of one's identity and providing role models from similar backgrounds can mitigate the effects of stereotype threat on female executives and improve their CSR–firm value performance.

# 7. Conclusions

This study examines the association between CSR and firm value. More notably, we investigate whether GP moderates the relationship between CSR and firm value. We also test whether gender plays a role in these observed associations.

First, our findings show that CSR is significantly and positively associated with firm value. Second, more interestingly, we find that the GP amplifies the positive association between CSR and firm value. We further tested to examine if gender plays a role in this observed association. Third, consistent with the general population's results, male CEOs with a golden parachute engage in more value-enhancing CSR projects than their counterparts without a golden parachute. In contrast with the results of the general population, women executives endowed with a GP do not perform well on the CSR–firm value assessment spectrum. The implication of this suggests that corporate CSR strategies should be adjusted based on the CEO's gender. Our results support both of our hypotheses.

As discussed above, in conclusion, when strategically implemented, for US-based firms, CSR could benefit society, firms, and corporate executives, thereby aligning the interests of these groups. This sheds propitious light and contributes to the CSR–firm value debate in related literature. Examining the moderating role of the GP is critical and peculiar to our study. The GP is here to stay, and US-based firms will continue to issue GPs to their executives [39]. So, findings ways where corporate practices critical to firms' success coexist and align with society's interests should be considered a welcome development.

Another important aspect of the findings in our study is the role that gender plays in the moderating impact of the GP on the CSR–firm value association. Several studies have demonstrated that gender diversity can significantly boost organizational innovation and firm performance [46,94–96]. However, firms must strategically promote an innovative culture that fuels success and growth by actively supporting gender diversity and developing inclusive workplaces where employees feel empowered and equally valued. Like their male counterparts, women should be equally acknowledged or given credit for their creative suggestions, encouraging them to always make decisions and contributions that maximize the firm's value.

Although we conducted extensive regression analyses and robustness tests to validate and strengthen our research results, this study has some limitations for which future studies may compensate. First, we used variables, including CSR and GP, that may be subject to measurement errors. Second, additional alternative and more extensive robustness tests could be employed to overcome the endogeneity issue in our regression analyses. One should be cautious in interpreting causal inferences in regression analyses. Third, our data are US-specific. Future research using non-US data could reveal even more interesting insights.

Despite this limitation, our study contributes two major practical implications. First, from the lens of social implications, the findings of this study show that society's interests and firms' long-term goals are achievable. As the head and driver of a firm's investment corporate policies, the CEO can pursue and engage in value-enhancing CSR that benefits both shareholders and the society from which they extract profit. Second, when designing corporate CSR strategies that align with shareholders' value, calculated adjustments based on gender tendencies should be made to better focus the CEO's incentive on CSR performance.

Author Contributions: Conceptualization, W.U.O.; Methodology, N.U.U. and E.C.; Writing–original draft, C.E.O., E.C. and K.L.J. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A.	Variable	Descriptions
-------------	----------	--------------

Variable	Definition
Dependent variable	S
TOBINQ	=(market value of equity + total liabilities)/book value of total assets, at fiscal year-end in period, respectively;
Variables of interest	
CSR	= summation of scores for community, diversity, employee relations, environment, human rights, and product in KLD's ratings;
GP	= a binary variable that equals 1 if the firms have adopted golden parachutes in that year and 0 otherwise;
Control variables	
SIZE	= natural logarithm of total assets at the end of the fiscal year;
LEVERAGE	= total liability divided by total assets at the end of the fiscal year;
GROWTH	= net sales minus net sales at the prior year and divided by the net sales at the end of the fiscal year;
ROA	= operating income divided by total assets at the end of the fiscal year;
RISK	= standard deviation of EBIT (earnings before interests, taxes) from the past five years divided by total assets at the end of the fiscal year;
R&D_INTENSITY	= R & D expenses divided by net sales at the end of the fiscal year;
GOVERNANCE	= net score (strengths minus concerns) in KLD governance ratings;
FIRM_AGE	= natural logarithm of $(1 + \text{the number of years since the firm's first appearance in the Computat database});$
CEO_AGE	= the age of the CEO;
CEO_TENURE	= how long the CEO has been serving in that capacity for the firm at the end of the fiscal year;
CEO_GENDER	= the gender of the CEO. A dummy variable that equals 1 if female and 0 if male;
IND	= industry indicator based on the two-digit SIC code;
YEAR	= year indicator.

# Appendix B. Description of KLD CSR Categories, Strengths, and Concerns

	Strengths			
Main Categories	Subcategories	Perfect Score	Actual Max.	Actual Mean
Community	<ul> <li>(1) Charitable Giving, (2) Innovative Giving, (3) Non-US Charitable</li> <li>Giving, (4) Support for Housing, (5) Support for Education,</li> <li>(6) Other Strength</li> </ul>	6	4	0.162
Corporate Governance	(1) Limited Compensation, (2) Ownership Strength, (3) Transparency Strength, (4) Political Accountability Strength, (5) Other Strength	5	3	0.143
Diversity	<ul> <li>(1) CEO, (2) Promotion, (3) Board of Directors, (4) Work/Life Benefits,</li> <li>(5) Employment of the Disabled, (6) Gay and Lesbian Policies,</li> <li>(7) Other Strength</li> </ul>	7	7	0.517
Employment	<ul> <li>(1) Union Relations, (2) No-Layoff Policy, (3) Cash Profit Sharing,</li> <li>(4) Employee Involvement, (5) Retirement Benefits Strength, (6) Health and Safety Strength, (7) Other Strength</li> </ul>	7	7	0.556
Environment	<ol> <li>Beneficial Products and Services, (2) Pollution Prevention,</li> <li>Recycling, (4) Clean Energy, (5) Communications, (6) Property, Plant, and Equipment, (7) Other Strength</li> </ol>	7	6	0.53
Human Rights	(1) Positive Record in South Africa, (2) Indigenous Peoples Relations Strength, (3) Labor Rights Strength, (4) Other Strength	4	2	0.037
Product	(1) Quality, (2) R&D/Innovation, (3) Benefits to Economically Disadvantaged, (4) Other Strength	4	3	0.147
	Total Score (Strength)	40	32	2.096
	Concerns			
Main Categories	Subcategories	Perfect Score	Actual Max.	Actual Mear
Community	<ul><li>(1) Investment Controversies, (2) Negative Economic Impact,</li><li>(3) Other Concern</li></ul>	3	3	0.061
Corporate Governance	<ul> <li>(1) High Compensation, (2) Ownership Concern, (3) Accounting Concern, (4) Transparency Concern, (5) Political Accountability,</li> <li>(6) Other Concern</li> </ul>	6	4	0.349
Diversity	(1) Controversies, (2) Non-Representation, (3) Other Concern	3	3	0.349
Employment	(1) Union Relations, (2) Health and Safety Concern, (3) Workforce Reductions, (4) Retirement Benefits Concern, (5) Other Concern	5	5	0.282

19 of 23

Concerns						
Main Categories	Subcategories	Perfect Score	Actual Max.	Actual Mean		
Environment	<ol> <li>Hazardous Waste, (2) Regulatory Problems, (3) Ozone Depleting Chemicals, (4) Substantial Emissions, (5) Agricultural Chemicals,</li> <li>(6) Climate Change, (7) Other Concern</li> </ol>	8	4	0.193		
Human Rights	<ul> <li>(1) South Africa, (2) Northern Ireland, (3) Burma, (4) Mexico, (5) Labor Rights Concern, (6) Indigenous Peoples Relations Concern,</li> <li>(7) Other Concern</li> </ul>	7	3	0.039		
Product	<ul><li>(1) Product Safety, (2) Marketing/Contracting Concern, (3) Antitrust,</li><li>(4) Other Concern</li></ul>	4	4	0.234		
	Total Score (Concern)	36	26	1.896		
	Source: KLD Research & Analytics Inc. (2006	<b>5</b> ).				

# References

- Cho, E.; Tsang, A. Corporate social responsibility, product strategy, and firm value. *Asia-Pac. J. Financ. Stud.* 2020, 49, 272–296. [CrossRef]
- 2. Graves, S.B.; Waddock, S.A. A look at the financial-social performance nexus when quality of management is held constant. *Int. J. Value-Based Manag.* **1999**, *12*, 87–99. [CrossRef]
- 3. Preston, L.E.; O'Bannon, D.P. The corporate social-financial performance relationship. Bus. Soc. 1997, 36, 419–429. [CrossRef]
- Ruf, B.M.; Muralidhar, K.; Brown, R.M.; Janney, J.J.; Paul, K. An empirical investigation of the relationship between change in corporate social performance and financial performance: A stakeholder theory perspective. *J. Bus. Ethics* 2001, 32, 143–156. [CrossRef]
- 5. Aupperle, K.E.; Carroll, A.B.; Hatfield, J.D. An empirical examination of the relationship between corporate social responsibility and profitability. *Acad. Manag. J.* **1985**, *28*, 446–463. [CrossRef]
- 6. Boyle, E.J.; Higgins, M.M.; Rhee, G.S. Stock market reaction to ethical initiatives of defense contractors: Theory and evidence. *Crit. Perspect. Account.* **1997**, *8*, 541–561. [CrossRef]
- 7. Brammer, S.; Jackson, G.; Matten, D. Corporate social responsibility and institutional theory: New perspectives on private governance. *Socio-Econ. Rev.* 2012, *10*, 3–28. [CrossRef]
- 8. Guerard, J.B., Jr. Is there a cost to being socially responsible in investing? J. Forecast. 1997, 16, 475–490. [CrossRef]
- 9. Moore, G. Corporate social and financial performance: An investigation in the UK supermarket industry. *J. Bus. Ethics* **2001**, *34*, 299–315. [CrossRef]
- 10. Frooman, J. Socially irresponsible and illegal behavior and shareholder wealth: A meta-analysis of event studies. *Bus. Soc.* **1997**, 36, 221–249. [CrossRef]
- 11. Clement-Jones, T. Bottom line issue or public relations exercise. Investing in Corporate Social Responsibility: A Guide to Best Practice, Business Planning and the UK's Leading Companies; Kogan Page: London, UK, 2005; pp. 5–13.
- Friedman, M. The social responsibility of business is to increase its profit. In New York Times Magazine; Springer: Berlin/Heidelberg, Germany, 1970; pp. 32–33.
- 13. Rudolph, P. Letters to the editor: An Adam Smith look at green regulations. *Wall Str. J.* 2005, 6.
- 14. Woolley, A.W.; Chabris, C.F.; Pentland, A.; Hashmi, N.; Malone, T.W. Evidence for a collective intelligence factor in the performance of human groups. *Science* **2010**, *330*, 686–688. [CrossRef]
- 15. Ely, R.J.; Ibarra, H.; Kolb, D.M. Taking gender into account: Theory and design for women's leadership development programs. *Acad. Manag. Learn. Educ.* **2011**, *10*, 474–493. [CrossRef]
- 16. McGuire, J.B.; Sundgren, A.; Schneeweis, T. Corporate social responsibility and firm financial performance. *Acad. Manag. J.* **1988**, 31, 854–872. [CrossRef]
- 17. El Ghoul, S.; Guedhami, O.; Kwok, C.C.Y.; Mishra, D.R. Does corporate social responsibility affect the cost of capital? *J. Bank. Financ.* **2011**, *35*, 2388–2406. [CrossRef]
- 18. Bae, K.-H.; Kang, J.-K.; Wang, J. Employee treatment and firm leverage: A test of the stakeholder theory of capital structure. *J. Financ. Econ.* **2011**, *100*, 130–153. [CrossRef]
- 19. Fombrun, C.; Shanley, M. What's in a name? Reputation building and corporate strategy. *Acad. Manag. J.* **1990**, *33*, 233–258. [CrossRef]
- 20. Margolis, J.D.; Elfenbein, H.A.; Walsh, J.P. Does it pay to be good? A meta-analysis and redirection of research on the relationship between corporate social and financial performance. *Ann. Arbor.* **2007**, *1001*, 48109-1234.
- 21. Wernerfelt, B. A resource-based view of the firm. *Strateg. Manag. J.* 1984, *5*, 171–180. [CrossRef]
- 22. Russo, M.V.; Fouts, P.A. A resource-based perspective on corporate environmental performance and profitability. *Acad. Manag. J.* **1997**, *40*, 534–559. [CrossRef]
- 23. McWilliams, A.; Siegel, D. Corporate social responsibility: A theory of the firm perspective. *Acad. Manag. Rev.* 2001, 26, 117–127. [CrossRef]
- 24. Hart, O. Corporate governance: Some theory and implications. Econ. J. 1995, 105, 678–689. [CrossRef]

- 25. Brown, T.J.; Dacin, P.A. The company and the product: Corporate associations and consumer product responses. *J. Mark.* **1997**, *61*, 68–84. [CrossRef]
- 26. Lev, B.; Petrovits, C.; Radhakrishnan, S. Is doing good good for you? How corporate charitable contributions enhance revenue growth. *Strateg. Manag. J.* 2010, *31*, 182–200. [CrossRef]
- 27. Turban, D.B.; Greening, D.W. Corporate social performance and organizational attractiveness to prospective employees. *Acad. Manag. J.* **1997**, 40, 658–672. [CrossRef]
- Luo, X.; Bhattacharya, C.B. Corporate social responsibility, customer satisfaction, and market value. J. Mark. 2006, 70, 1–18. [CrossRef]
- 29. Gardberg, N.A.; Fombrun, C.J. Corporate citizenship: Creating intangible assets across institutional environments. *Acad. Manag. Rev.* **2006**, *31*, 329–346. [CrossRef]
- Godfrey, P.C. The relationship between corporate philanthropy and shareholder wealth: A risk management perspective. *Acad. Manag. Rev.* 2005, 30, 777–798. [CrossRef]
- Goss, A.; Roberts, G.S. The impact of corporate social responsibility on the cost of bank loans. J. Bank. Financ. 2011, 35, 1794–1810. [CrossRef]
- 32. Barnea, A.; Rubin, A. Corporate social responsibility as a conflict between shareholders. J. Bus. Ethics 2010, 97, 71–86. [CrossRef]
- Cai, Y.; Jo, H.; Pan, C. Vice or virtue? The impact of corporate social responsibility on executive compensation. *J. Bus. Ethics* 2011, 104, 159–173. [CrossRef]
- 34. Mahoney, L.S.; Thorn, L. An examination of the structure of executive compensation and corporate social responsibility: A Canadian investigation. *J. Bus. Ethics* **2006**, *69*, 149–162. [CrossRef]
- 35. Harjoto, M.A.; Jo, H. Corporate governance and CSR nexus. J. Bus. Ethics 2011, 100, 45–67. [CrossRef]
- Cespa, G.; Cestone, G. Corporate social responsibility and managerial entrenchment. J. Econ. Manag. Strategy 2007, 16, 741–771. [CrossRef]
- 37. Gul, F.A.; Kim, J.-B.; Qiu, A.A. Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: Evidence from China. *J. Financ. Econ.* **2010**, *95*, 425–442. [CrossRef]
- 38. Ujah, N.U.; Okafor, C.E. The effect of a golden parachute on managed earnings. Manag. Financ. 2019, 45, 925–949. [CrossRef]
- 39. Okafor, C.E.; Ujah, N.U. Executive compensation and corporate social responsibility: Does a golden parachute matter? *Int. J. Manag. Financ.* **2020**, *16*, 575–598. [CrossRef]
- 40. Mahoney, L.S.; Thorne, L. Corporate social responsibility and long-term compensation: Evidence from Canada. *J. Bus. Ethics* 2005, 57, 241–253. [CrossRef]
- 41. Kane, E.J. Using deferred compensation to strengthen the ethics of financial regulation. *J. Bank. Financ.* **2002**, *26*, 1919–1933. [CrossRef]
- 42. Mahapatra, S. Investor reaction to a corporate social accounting. J. Bus. Financ. Account. 1984, 11, 29–40. [CrossRef]
- 43. Dezsö, C.L.; Ross, D.G. Does female representation in top management improve firm performance? A panel data investigation. *Strateg. Manag. J.* **2012**, *33*, 1072–1089. [CrossRef]
- 44. Noland, M.; Moran, T.; Kotschwar, B.R. *Is Gender Diversity Profitable? Evidence from a Global Survey*; Working Paper 16-3; Peterson Institute for International Economics: Washington, DC, USA, 2016.
- 45. Hewlett, S.A.; Marshall, M.; Sherbin, L. How diversity can drive innovation. Harvard Bus. Rev. 2013, 91, 30.
- 46. Hunt, V.; Layton, D.; Prince, S. Diversity matters; McKinsey & Company: Atlanta, GA, USA, 2015; Volume 1, pp. 15–29.
- 47. Huddy, L.; Terkildsen, N. Gender stereotypes and the perception of male and female candidates. *Am. J. Political Sci.* **1993**, 37, 119–147. [CrossRef]
- Bass, B.M.; Avolio, B.J. Shatter the glass ceiling: Women may make better managers. *Hum. Resour. Manag.* 1994, 33, 549–560.
   [CrossRef]
- 49. Swann, W.B., Jr. Self-verification theory. In Handbook of Theories of Social Psychology; Sage Publications Ltd.: London, UK, 2012.
- 50. Eagly, A.H. Sex Differences in Social Behavior: A Social-Role Interpretation; Psychology Press: London, UK, 2013.
- Schmader, T.; Johns, M. Converging evidence that stereotype threat reduces working memory capacity. J. Personal. Soc. Psychol. 2003, 85, 440–452. [CrossRef] [PubMed]
- 52. Nguyen, H.-H.D.; Ryan, A.M. Does stereotype threat affect test performance of minorities and women? A meta-analysis of experimental evidence. *J. Appl. Psychol.* 2008, 93, 1314–1334. [CrossRef] [PubMed]
- Steele, C.M.; Aronson, J. Stereotype threat and the intellectual test performance of African Americans. J. Personal. Soc. Psychol. 1995, 69, 797. [CrossRef]
- 54. Inzlicht, M.; Kang, S.K. Stereotype threat spillover: How coping with threats to social identity affects aggression, eating, decision making, and attention. *J. Personal. Soc. Psychol.* **2010**, *99*, 467–481. [CrossRef]
- 55. Good, C.; Aronson, J.; Inzlicht, M. Improving adolescents' standardized test performance: An intervention to reduce the effects of stereotype threat. *J. Appl. Dev. Psychol.* **2003**, *24*, 645–662. [CrossRef]
- 56. Adams, R.B.; Funk, P. Beyond the glass ceiling: Does gender matter? Manag. Sci. 2012, 58, 219–235. [CrossRef]
- 57. Deckop, J.R.; Merriman, K.K.; Gupta, S. The effects of CEO pay structure on corporate social performance. *J. Manag.* 2006, 32, 329–342. [CrossRef]
- 58. Szwajkowski, E.; Figlewicz, R.E. Evaluating corporate performance: A comparison of the Fortune reputation survey and the Socrates social rating database. *J. Manag. Issues* **1999**, *11*, 137–154.

- 59. Waddock, S.A.; Graves, S.B. The corporate social performance–financial performance link. *Strateg. Manag. J.* **1997**, *18*, 303–319. [CrossRef]
- 60. Mattingly, J.E.; Berman, S.L. Measurement of corporate social action discovering taxonomy in the Kinder Lydenburg Domini ratings data. *Bus. Soc.* 2006, 45, 20–46. [CrossRef]
- 61. Chatterji, A.K.; Levine, D.I.; Toffel, M.W. How well do social ratings actually measure corporate social responsibility? *J. Econ. Manag. Strategy* **2009**, *18*, 125–169. [CrossRef]
- 62. Dhaliwal, D.S.; Li, O.Z.; Tsang, A.; Yang, Y.G. Voluntary nonfinancial disclosure and the cost of equity capital: The initiation of corporate social responsibility reporting. *Account. Rev.* **2011**, *86*, 59–100. [CrossRef]
- 63. Kim, J.; Choi, D.; Cho, E.; Okafor, C.E.; Park, B.I. Corporate Sustainability in the Wake of the COVID-19 Global Pandemic: Does CSR Enhance Corporate Survival during a Market Crisis? *Sustainability* **2022**, 14, 14438. [CrossRef]
- 64. Kim, Y.T.; Park, M.S.; Wier, B. Is earnings quality associated with corporate social responsibility? *Account. Rev.* **2012**, *87*, 761–796. [CrossRef]
- 65. Cho, E.H.; Chun, S.B. Corporate social responsibility, real activities earnings management, and corporate governance: Evidence from Korea. *Asia-Pac. J. Account. Econ.* **2016**, *23*, 400–431. [CrossRef]
- 66. Wang, H.; Choi, J.; Li, J. Too little or too much? Untangling the relationship between corporate philanthropy and firm financial performance. *Organ. Sci.* 2008, *19*, 143–159. [CrossRef]
- 67. Lien, Y.C.; Li, S. Does diversification add firm value in emerging economies? Effect of corporate governance. J. Bus. Res. 2013, 66, 2425–2430. [CrossRef]
- Servaes, H.; Tamayo, A. The impact of corporate social responsibility on firm value: The role of customer awareness. *Manag. Sci.* 2013, 59, 1045–1061. [CrossRef]
- 69. Choi, D.; James, K.L.; Okafor, C.E.; Cho, E.; Bhuyan, M.N.H. The effect of executives' education and compensation on firm value: Evidence from China. *Appl. Econ.* **2022**, *54*, 5763–5781. [CrossRef]
- 70. Oh, W.; Chang, Y.; Martynov, A. The effect of ownership structure on corporate social responsibility: Empirical evidence from Korea. J. Bus. Ethics 2011, 104, 283–297. [CrossRef]
- 71. Seifert, B.; Morris, S.A.; Bartkus, B.R. Having, giving, and getting: Slack resources, corporate philanthropy, and firm financial performance. *Bus. Soc.* **2004**, *43*, 135–161. [CrossRef]
- 72. Zhang, R.; Zhu, J.; Yue, H.; Zhu, C. Corporate philanthropic giving, advertising intensity, and industry competition level. *J. Bus. Ethics* **2010**, *94*, 39–52. [CrossRef]
- McWilliams, A.; Siegel, D. Corporate social responsibility and financial performance: Correlation or misspecification? *Strateg. Manag. J.* 2000, 21, 603–609. [CrossRef]
- 74. Mishra, D.R. Post-innovation CSR performance and firm value. J. Bus. Ethics 2017, 140, 285–306. [CrossRef]
- 75. Joh, S.W. Corporate governance and firm profitability: Evidence from Korea before the economic crisis. *J. Financ. Econ.* **2003**, *68*, 287–322. [CrossRef]
- 76. Baek, J.; Kang, J.; Park, S. Corporate governance and firm value: Evidence from the Korean financial crisis. *J. Financ. Econ.* **2004**, 71, 265–313. [CrossRef]
- 77. Drobetz, W.; Schillhofer, A.; Zimmermann, H. Corporate governance and expected stock returns: Evidence from Germany. *Eur. Financ. Manag.* **2004**, *10*, 267–293. [CrossRef]
- Black, B.S.; Jang, H.; Kim, W. Does corporate governance predict firms' market values? Evidence from Korea. J. Law Econ. Organ. 2006, 22, 366–413. [CrossRef]
- 79. Cline, B.N.; Yore, A.S. Silverback CEOs: Age, experience, and firm value. J. Empir. Financ. 2016, 35, 169–188. [CrossRef]
- Dikolli, S.S.; Mayew, W.J.; Nanda, D. CEO tenure and the performance-turnover relation. *Rev. Account. Stud.* 2014, 19, 281–327. [CrossRef]
- Brochet, F.; Limbach, P.; Schmid, M.; Scholz-Daneshgari, M. CEO tenure and firm value CEO tenure and firm value. *Account. Rev.* 2021, 96, 47–71. [CrossRef]
- 82. Ullah, I.; Fang, H.; Jebran, K. Do gender diversity and CEO gender enhance firm's value? Evidence from an emerging economy. *Corp. Gov. Int. J. Bus. Soc.* **2019**, 20, 44–66. [CrossRef]
- 83. Bebchuk, L.; Cohen, A.; Wang, C.C. Golden parachutes and the wealth of shareholders. *J. Corp. Financ.* 2014, 25, 140–154. [CrossRef]
- 84. Bebchuk, L.; Cohen, A.; Ferrell, A. What matters in corporate governance? Rev. Financ. Stud. 2009, 22, 783–827. [CrossRef]
- 85. Lawrence, A.; Minutti-Meza, M.; Ping, Z. Can big 4 versus non-big 4 differences in audit-quality proxies be attributed to client characteristics? *Account. Rev.* **2011**, *86*, 259–286. [CrossRef]
- 86. Masulis, R.W.; Wang, C.; Xie, F. Corporate governance and acquirer returns. J. Financ. 2007, 62, 1851–1889. [CrossRef]
- 87. Kim, Y.; Wee, J. Comprehensive analysis of corporate social responsibility. *Korean J. Bus. Adm.* **2011**, *24*, 2913–2950.
- Jiraporn, P.; Jiraporn, N.; Boeprasert, A.; Chang, K. Does corporate social responsibility (CSR) improve credit ratings? Evidence from geographic identification. *Financ. Manag.* 2014, 43, 505–531. [CrossRef]
- 89. Harjoto, M.A.; Jo, H. Legal vs. normative CSR: Differential impact on analyst dispersion, stock return volatility, cost of capital, and firm value. *J. Bus. Ethics* 2015, 128, 1–20. [CrossRef]
- Zaefarian, G.; Kadile, V.; Henneberg, S.C.; Leischnig, A. Endogeneity bias in marketing research: Problem, causes and remedies. *Ind. Mark. Manag.* 2017, 65, 39–46. [CrossRef]

- Porter, M.E.; Kramer, M.R. Creating shared value. In *Managing Sustainable Business*; Springer: Berlin/Heidelberg, Germany, 2019; pp. 323–346.
- 92. Barney, J. Firm resources and sustained competitive advantage. J. Manag. 1991, 17, 99–120. [CrossRef]
- 93. Mogavero, D.J.; Toyne, M.F. The impact of golden parachutes on Fortune 500 stock returns: A reexamination of the evidence. *Q. J. Bus. Econ.* **1995**, *34*, 30–38.
- 94. Smith, N.; Smith, V.; Verner, M. Do women in top management affect firm performance? A panel study of 2500 Danish firms. *Int. J. Product. Perform. Manag.* **2006**, *55*, 569–593. [CrossRef]
- 95. Herring, C. Does diversity pay? Race, gender, and the business case for diversity. Am. Sociol. Rev. 2009, 74, 208–224. [CrossRef]
- Catalyst, Why Diversity and Inclusion Matter. 2018. Available online: www.catalyst.org/research/why-diversity-and-inclusionmatter (accessed on 12 March 2023).

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.